

## **Author's response to reviews**

**Title:** Exploring the Association Between Morgellons Disease and Lyme Disease: Identification of *Borrelia burgdorferi* in Morgellons Disease Patients.

### **Authors:**

Marianne J. Middelveen ([middel@telus.net](mailto:middel@telus.net))  
Cheryl Bandoski ([cband1@unh.newhaven.edu](mailto:cband1@unh.newhaven.edu))  
Jennie Burke ([Jennie.burke@australianbiologics.com.au](mailto:Jennie.burke@australianbiologics.com.au))  
Eva Sapi ([unh@evasapi.net](mailto:unh@evasapi.net))  
Katherine R. Filush ([KFilu1@unh.newhaven.edu](mailto:KFilu1@unh.newhaven.edu))  
Yean Wang ([techinfo@australianbiologics.com.au](mailto:techinfo@australianbiologics.com.au))  
Agustin Franco ([agustin.franco@optusnet.com.au](mailto:agustin.franco@optusnet.com.au))  
Peter J. Mayne ([quickpeter@gmail.com](mailto:quickpeter@gmail.com))  
Raphael B. Stricker ([rstricker@usmamed.com](mailto:rstricker@usmamed.com))

**Version:2****Date:**25 August 2014

**Author's response to reviews:** see over



## International Lyme and Associated Diseases Society

P.O. Box 341461 • Bethesda, Maryland 20827-1461

[www.ilads.org](http://www.ilads.org)

E-mail: [lymedocs@aol.com](mailto:lymedocs@aol.com)

August 24, 2014

Maria Merrie Jul Ladag  
Journal Editorial Office  
BioMed Central

**Re: MS: 9014434614030845 - Revisions required prior to editorial assessment**

Dear MJ,

We have revised the above manuscript as you requested. Specifically we have done the following:

1. Competing Interests: We have included a “Competing Interests” section between the Conclusions and Authors' Contributions sections.
2. Authors' Contributions: We have included an “Authors' Contributions” section before the Acknowledgements and Reference list.
3. Consent: We have stated in the Methods section that written informed consent for participation in the study was obtained from all participants.
4. Line Numbering: We have revised the manuscript to include line and page numbers.
5. Title Page: The Title Page has been revised as requested.

We appreciate your help with the manuscript processing for *BMC Dermatology*.

Sincerely,

Raphael Stricker, MD  
President-Elect, ILADS